Author's response to reviews

Title: Effective antiprotease-antibiotic treatment of experimental anthrax

Authors:

Serguei G Popov (spopov@gmu.edu)
Taissia G Popova (tpopova@gmu.edu)
Svetlana Hopkins (svetlana.hopkins@analex.com)
Raymond S Weinstein (rweinste@gmu.edu)
Rebecca MacAfee (rmacafee@potomach.com)
Karl J Fryxell (kfryxell@gmu.edu)
Vikas Chandhoke (vchandho@gmu.edu)
Charles Bailey (cbailey2@gmu.edu)
Ken Alibek (kalibek@gmu.edu)

Version: 6 Date: 6 April 2005

Author's response to reviews:

1. Affiliation country has been added to #2
2. Table has been moved to after figure legends
3. Spacing in the first row of the table has been adjusted